Author’s response to reviews

Title: The Care Process Self-Evaluation Tool: a valid and reliable tool for health care teams to measure the level of organization of care processes

Authors:

Deborah Seys (deborah.seys@med.kuleuven.be)
Svin Deneckere (svin.deneckere@med.kuleuven.be)
Walter Sermeus (walter.sermeus@med.kuleuven.be)
Eva Van Gerven (eva.vangerven@med.kuleuven.be)
Massimiliano Panella (panella@med.unipmn.it)
Luk Bruyneel (luk.bruyneel@med.kuleuven.be)
Timothy Mutsvari (timothy.mutsvari@med.kuleuven.be)
Rafaela Camacho Bejarano (rafaela.camacho@denf.uhu.es)
Seval Kul (sevalkul@gantep.edu.tr)
Kris Vanhaecht (kris.vanhaecht@med.kuleuven.be)

Version: 2 Date: 17 September 2012

Author’s response to reviews: see over
Submission for publication of the manuscript ‘The care process self-evaluation tool: a valid and reliable tool for health care teams to measure the level of organization of care processes’

Dear Editor

We apologize that the competing interests, ethics statement and title page was not included in the previous version of the manuscript.
The authors declare that they have no competing interests. This is placed after the conclusion section.
The ethical approval was obtained as part of a larger research project. The study protocol of this has been published in Implementation Science and all necessary documents have been made available to Biomed Central. (Deneckere S, Euwema M, Lodewijckx C, Panella M, Sermeus W, Vanhaecht K: The European quality of care pathways (EQCP) study on the impact of care pathways on interprofessional teamwork in an acute hospital setting: study protocol: for a cluster randomised controlled trial and evaluation of implementation processes. Implement Sci 2012, 7: 47 doi: 10.1186/1748-5908-7-47). The ethical approval is added in the methods section.

Best regards,

Dr. Kris Vanhaecht, Leuven University, Belgium (corresponding author)